Review

Causal Impacts of the COVID-19 Pandemic on Food Supply Chains: A Systematic Review

Brenda Cardoso¹, Luiza Cunha¹, Adriana Leiras¹, Paulo Gonçalves², Hugo Yoshizaki³, Irineu de Brito Junior⁴, Frederico Pedroso⁵

- ¹ Industrial Engineering Department, Pontifical Catholic University of Rio de Janeiro PUC-Rio, Rio de Janeiro, Brazil; <u>brendafarias.eng@outlook.com</u> (B.C.); <u>luizarac@gmail.com</u> (L.C); <u>adrianaleiras@puc-rio.br</u> (A.L.)
- ² Facoltà di Scienze Economiche, Università della Svizzera Italiana, Lugano, TI, Switzerland; <u>paulo.goncalves@usi.ch</u> (P.G.)
- ³ Production Engineering, Poli, Sao Paulo University, Sao Paulo, Brazil; <u>hugo@usp.br</u> (H.Y.)
- ⁴ Environmental Engineering Department, Sao Paulo State University, Sao Jose dos Campos, Brazil; <u>ir-ineu.brito@unesp.br</u> (I.B.J.)
- ⁵ Urban, Disaster Risk Management, Resilience & Land, World Bank; fpedroso@worldbank.org (F.P.)
- * Correspondence: adrianaleiras@puc-rio.br; Tel.: +55 (21) 3527 1285/1286

Abstract: The COVID-19 pandemic has severely affected the food supply chain, including producers, retailers, wholesalers, and customers. To minimize the impacts caused by pandemics and epidemics on food supply chains, it is fundamental to implement effective policies that ensure continuity in the provision, affordability, and distribution of basic food items. This research aims to identify the main impacts of pandemics and epidemics on food supply chains and policies that can minimize these impacts. Based on a systematic literature review (SLR), 174 documents are analysed to propose a taxonomy of impacts on four supply chain links: demand-side, supply-side, logistics and infrastructure, and management and operation. The taxonomy presents the main impacts, as well as the respective mitigation policies simultaneously. In addition, the literature review leads to the development of a comprehensive causal loop diagram (CLD) with the identification of main variables and their relationship with food supply chains. Finally, a specific research agenda is proposed by identifying main research gaps. These findings provide a structured method for evaluating policies that ensure the functioning of food supply chains, particularly in disruptions such epidemics and pandemics.

Keywords: COVID-19; Food Supply Chain; Epidemic; Pandemic, Disruptions

1. Introduction

Disasters can trigger Supply Chain (SC) disruptions [1]. SCs operate in a level of high uncertainty during disaster response, which is very different from what most SC managers experience in regular operations [2]. During disasters, for example, companies face challenges such as time pressure, increased information uncertainty, and dynamic changes over time [3]. Besides, coordination among suppliers is difficult at best, which may disrupt SC flows.

Disasters demand efforts on Food Supply Chains' (FSCs) resilience [4,5]. FSCs refer to all processes involved for food reaching the final consumer, including cultivation, processing, distribution, sales, and consumption [5]. FSCs need to operate in a relatively delicate balance between consumption, production, and inventory. Any disturbance in FSCs can cause instability due to delays between decisions and results, introducing oscillation and amplification [6,7]. Several factors can disrupt food supply chains, including political instability, droughts, floods, and infectious diseases [7,8]. FSCs are particularly susceptible to disruption because they are designed for efficiency and not for flexibility [9]. Most recently, the COVID-19 pandemic is significantly impacting FSCs [10,11,12,13,14], as breakdowns or bottlenecks in some elements have impacted other components up and down their chains [11]. These increased impacts pose significant new challenges to organizations, requiring better designed and more effective inter-organizational collaboration at national and regional levels. A major challenge for companies during a pandemic is to implement systems (e.g., food production) resilient enough to continue functioning during a disaster [15]. To maintain the sustainability of their business, it is necessary to implement organizational practices [16].

Thus, this research aims to present the main impacts caused by pandemics and epidemics on FSCs and the related policies to mitigate their severity. Therefore, we seek to shed light on the following research questions:

- What are the main impacts of pandemics and epidemics on food supply chains?
- Which policies can effectively mitigate the impacts of pandemics and epidemics on food supply chains?

This paper involves an exhaustive compilation of data and information through a Systematic Literature Review (SLR), collected from scientific documents and grey literature on pandemics and epidemics and the FSCs. We deliver three SLR possible outcomes proposed by [17]: a taxonomy, a framework, and a research agenda. The taxonomy, represented by tables evidencing impacts and policies, classifies the extant literature. The conceptual framework, illustrated through a causal loop diagram (CLD), synthesizes the literature based on our critical assessment. The research agenda poses propositions for new research [17].

Since the global food system is complex and the coupled relationships between different variables and effects are dynamic and nonlinear, it is critical to use suitable analytical modelling techniques, such as System Dynamics (SD). SD captures the dynamic complexity and the nonlinear behavior of complex systems over time and provides ways to understand the system behavior through feedback mechanisms [6]. Thus, SD enables a structured analysis of the food supply chain and policies to address the impact of the COVID-19 pandemic.

This study is relevant given the challenges faced by decision-makers in the context of pandemics and epidemics, in addition to the detrimental social impact that the subject involves. This paper builds upon a research gap on the impacts of pandemics and epidemic (e.g., COVID-19) on supply chains that are not yet sufficiently discussed [1,12]. Previous research has discussed fragmented policies for the functioning of FSCs during crisis and emergency times; however, pandemics and epidemics need attention to the development of potential integrated policies to mitigate the risks of FSCs. Therefore, this paper brings a synthesis of academic literature related to the pandemics and epidemics' impacts and risk mitigation policies.

Following this introduction, Section 2 presents the research methodology. Section 3 reports the SLR and grey literature analysis, including a taxonomy and a framework (causal loop diagram). Section 4 presents a discussion, theoretical and practical implications, and a research agenda. Finally, Section 5 summarizes the concluding remarks, limitations, and future research avenues.

2. Materials and Methods

This research employs the method of Systematic Literature Review (SLR) through an eight-step process, as detailed in [18]: (i) planning and problem formulation, (ii) literature search, (iii) data gathering, (iv) quality evaluation, (v) data analysis and synthesis, (vi) interpretation, (vii) presenting the results, and (viii) review updating.

The research problem formulation has the two research questions presented in the Introduction. The second step encompasses a bibliographic search in both Scopus and Web of Science (WoS) databases, as they have a significant number of indexed journals [19]. The search considers three groups of keywords defined by a combination of keywords that covers the topic broadly enough to avoid any artificial limitation of the documents obtained while providing limits to exclude undesirable results [20]. The first group of keywords addressed the research SC focus. The second group includes impact terminologies. The third group of keywords encompasses pandemic and epidemic terminologies. The set of keywords is as follows: ("suppl* chain*") AND (food) AND (impact* OR disruption* OR security OR safety OR effect* OR consequence* OR break* OR rupture* OR influenc*) AND (epidemic* OR pandemic* OR covid* OR corona* OR influenza OR sars OR mers OR ebola OR zika OR chikungunya)). The SLR, carried out on March 29, 2021, returned 534 documents without initial exclusions. The documents were analysed according to the following inclusion and exclusion criteria:

- Inclusion criteria: Research on the impact of pandemics and epidemics on FSCs, or research with proposed solutions for disruption in supply chains caused by pandemics or epidemics.
- Exclusion criteria: Research outside the context of pandemics and epidemics; research that does not address FSCs, or that addresses chain disruption for food poisoning.

Figure Figure 1 summarizes the steps using the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) diagram [21].





The stage of data collection is performed by employing auxiliary spreadsheets with the essential elements (impacts, policies, and strategies) and also categories of documents according to their research application [22,23].

The detailed description of the SLR ensured the quality evaluation stage. Also, each document has been reviewed independently by two reviewers. The agreement index between the reviewers was 94% when reading the abstracts, thus being considered acceptable [24]. The data analysis and synthesis are presented in the next section, describing the main results obtained from the documents' complete reading. Still, the paper synthesis is delivered through a taxonomy and framework (causal loop diagram) with the leading

SLR results. The discussion is based on how impact evaluation is addressed, considering a food supply chain perspective. The presentation of the results is described in the present paper. Updating of the SLR is proposed as future research.

In addition to the peer-reviewed literature, this study also considers reports, news and sites – what is called grey literature. The search also considers the topic of pandemics and epidemics impact in food supply chains. It is worth mentioning that the grey literature review is carried out in the context of the COVID-19 pandemic since it is a current disaster at the time of this research. Therefore, to perform the search in grey literature, the Google platform was used in the advanced search option. In this way, the query used for the search regarding the FSCs is: ('covid-19 OR coronavirus' 'impact OR disruption' food "supply chain"). The search was carried in three major international sites (BBC News World, CNN News World, and Reuters) and considered documents from March 11, 2020 (the date on which WHO declared the COVID-19 pandemic) until March 29, 2021.

3. Literature Analysis

The following sections present descriptive and content analysis that lead to taxonomies for the main impacts of pandemics and epidemics in food supply chains, as well as respective mitigation policies. Additionally, this section also presents our proposed framework (causal loop diagram).

3.1. Descriptive Analysis

Figure Figure **2** shows the evolution of documents per year, the first one appearing in 2009. In the years 2009 and 2010 the world faced the H1N1 flu pandemic, leading to the appearance of publications in this field of study. The peak is reached in 2020 (61 documents) due to the recent coronavirus pandemic.



Figure 2. Evolution of publications per year until March, 2021 (academic literature).

Regarding the type of publication of the SLR documents (academic literature), 80% are articles published in academic journals; of these, 11% are review papers, and the other documents are divided between papers published in conference proceedings, book chapters, and notes.

Concerning publication channels of the SLR documents (academic literature), Table Table 1 presents the SLR documents by periodic/journal. *Sustainability* stands out with nine publications, followed by the *Canadian Journal of Agricultural Economics*, which has seven works on the topic. *Global Food Security, Agricultural Systems*, and *EuroChoices* have five papers each. Those publications are inserted in journals related to sustainability, business, nutrition, development, and economics, which characterize the theme as multidisciplinary.

Periodic/Journal	Quantity
Sustainability (Switzerland)	9
Canadian Journal of Agricultural Economics	7
Global Food Security	5
Agricultural Systems	5
EuroChoices	5
China Agricultural Economic Review	4
Food Control	3
World Development	3
Applied Economic Perspectives and Policy	3
International Journal of Environmental Research and Public Health	3
Food Security	3
Others	60

Table 1. Publication channels (academic literature)

3.2. Content Analysis

Previous literature recognizes the importance of analyzing impacts and vulnerabilities of FSCs to mitigate risks of interruption. Identifying the various ways food systems may fail is fundamental so that stakeholders, especially government officials, may prioritize actions and plans accordingly [25].

Content analysis is performed by fully reading each document and categorizing their data. Categorization, in turn, is an essential tool for systematic analysis of documents, based on an iterative process with a constant interplay of tests, reviews, and data comparison. Thus, this paper considers the impacts and policies from each document (Appendix Appendix A – Impacts of pandemics and epidemics and respective mitigation policies on FSC). The classification of these impacts and policies results in categories of FSCs in the context of pandemics and epidemics (Table Table 2).

Impacts are grouped into nine categories: human resources constraints; SC instability; financial and economic constraints; logistics disruption; infrastructure disruption; relationships between stakeholders; difficulty for accessing services and materials; consumer behavior; and unemployment and reduced income. Policies are grouped into categories according to their associated impact.

Additionally, this study identifies the affected SC link: supply-side; demand-side; logistic and infrastructure; and management and operation. Food supply chains had to adjust rapidly to demand-side shocks and also plan for any supply-side disruptions [26]. Moreover, it is necessary to define strategies for impacts on logistics and infrastructure, and management and operation perspectives [27].

Table 2. Impacts and policies (summary)

		S	C Link		
Impact category	Supply- side	Demand- side	Logistic and infrastructure	Management and operation	Policy category
Human Resources constraints	Х		х	Х	Workforce assessment and monitoring Workforce adjustment Activities automation Health Safety Protocols
SC instability	Х	x	Х	X	Stability and business continuity plans Government financial assistance for companies Cooperation and collaboration Online infrastructures Decision and communication support tools
Financial and economic con- straints	х		х	Х	Economic and reduce costs plans Government awereness of economic impacts
Logistics disruption	Х		Х	Х	Decision and communication support tools Government financial assistance for companies Online infrastructures Transportation and distribution plans
Infrastructure disruption	X		Х	Х	Restructuring plans Online infrastructures Workforce adjustment
Relationships between stake- holders	х		Х	Х	Communication technologies Cooperation and collaboration
Difficulty accessing services and materials	Х		Х	х	Government financial assistance for companies Contingency plans Cooperation and collaboration
Consumer behavior	Х	Х		Х	Demand analysis plans Information sharing tools
Unemployment and reduced income		x			Social support programs Food security plans Government financial assistance for population Workforce adjustment

2

One of the significant impacts of pandemics and epidemics on the FSCs is on human 3 resources constraints. In the COVID-19 pandemic, specifically, staffing shortages (not 4 enough workers) has become a significant problem in several countries [28]. Restrictive 5 movement measures (e.g., lockdown) caused labor shortages, which affect activities in the 6 food supply chain, harvesting, processing, and distribution [8]. Consequently, as the 7 availability of human resources is lower, the disruption in the normal flow of activities in 8 the food supply chain increases. Cargo transportation companies (e.g., airfreight) and the 9 agriculture sector are links severely impacted by the lack of personnel [28,29,30]. 10

SC instability is another impact highlighted by some authors. During the COVID-19 11 pandemic, oscillation of food prices can be observed [11,12,13,31]. Price increases might 12 be partially due to supply-side disruptions [32,33]. Trade frictions could also affect global 13 food prices [34]. Some authors still address the impact of instability in food supply and 14 demand during pandemics and epidemics [5,35,36]. Supply and demand fluctuation, in 15 general, can trigger instability of inventory levels [37] since most organizations are not 16 prepared to operate in crisis and emergency scenarios. Consequently, companies in all 17 sectors are targets of financial losses and increases in operating costs [35,36,38,39,40,41], 18 adverse effects for business growth and productivity [39,42,43], delays and interruptions 19 for the delivery of products, and problems in packaging materials [28]. 20

Financial and economic constraints also have significant impacts on FSCs. During 21 pandemics and epidemics such as the COVID-19, companies may suffer from lack of 22 working capital, given that operational costs have increased [44,45]. 23

Logistics and infrastructure impacts are also highly cited (e.g., [41,46,47,48,49,50]). 24 Disruption in transportation, for example, can be concisely circumscribed to the unavail-25 ability of labor, since often laborers need to stay at home and are unavailable to carry out 26 their activities, due to social distancing policies or infected workers [26,51,52]. Movement 27 restrictions are also considered an impact since some countries block or restrict the export 28 of basic products to guarantee national food security [53]. From this perspective, it is also 29 possible to identify that the occurrence of pandemics and epidemics affects food distribu-30 tion [54,55,56]. Once a pandemic or epidemic affects the trade of goods (e.g., additional 31 border controls; export restrictions; import restrictions), product movement is also a 32 highly mentioned supply-side impact [34,57,58,59]. The interruption in supply chain 33 movements result in a cascading effect on producers and traders [60]. Besides, partial or 34 total blocking measures resulted in the closure of schools, universities, workplaces, shops 35 and non-essential restaurants. Those restrictions changed means to food access and served 36 as another obstacle to distributing food to populations [56]. Some companies had to close 37 for cleaning purposes [61], while others (e.g., slaughterhouses) had to close due to the 38 high number of infected people. 39

Some authors mention the relationship between stakeholders as an impact during 40pandemics and epidemics. During a disaster, organizations face challenges, for example, 41 related to great information uncertainty [62], which hamper communications between 42 links in the supply chain. In addition, as pandemics hit supply chain infrastructure, busi-43 ness relationships can be undermined [44]. The difficulty in accessing equipment, services, 44 and materials also impacts FSCs, especially for producers. The COVID-19 pandemic, for 45 example, affected the agricultural sector, disrupting the supply of inputs (e.g., fertilizers, 46 seeds, machinery, workers, and extension services) thus affecting production systems and 47 supply chains [63,64]. 48

The change in consumer behavior during pandemics and epidemics also interfere 49 negatively on FSCs, especially on the types of products purchased. During a disaster such 50 as the COVID-19 pandemic, people become cautious about their spending, buying less 51 often but in larger volumes, preferring small grocery stores, and purchasing food online 52 to avoid agglomerations [13,65,66]. The initial peak in demand due to panic buying behaviors was also a problem [67]. Panic buying of household items and store cupboard 54 staples such as rice and flour has occurred in nearly every country hit by the virus [28]. 55 Panic buying behavior typically originates from customers who buy more than usual and not restricted food availability. This trend was partially driven by the media, which often showed pictures of empty shelves and warned of food shortages [69]. It also affected people from all income classes, including lower income populations [70].

Finally, with the occurrence of pandemics and epidemics, family income may be lower than adequate due to temporary dismissal or permanent unemployment of a member [71]. The COVID-19 pandemic, for example, resulted in an economic slowdown with consequent job losses and decreased income [72,73]. 63

All those impacts, as mentioned earlier, contribute to food insecurity of the vulnera-64 ble population. Food security is a global concern and is a challenge for many middle- and 65 lower-income countries [73]. The economic slowdown caused by COVID-19, with conse-66 quent job losses, decreased income, and disruption in SC, seriously threatens food security 67 [72]. Therefore, food systems need to adopt robust strategies and policies to correlate the 68 needs of producers, farmers, workers, low-income consumers, distribution, and the food 69 chain as a whole [41,57]. Strategies and policies are essential to social measures, once they 70 must guarantee income access for all families and actions designed to protect vulnerable 71 populations. Among the proposed policies, the following stand out: social protection and 72 food programs [5,67]; measures as social assistance and incentive [25,67]; and, aid and 73 food donations [57,72]. 74

Based on the main impacts and policies found in the SLR, we developed a framework (CLD) to identify main cause and effect relationships between the variables. 76

3.3. Framework (Causal loop diagram)

Figure Figure 3. CLD (Framework) with variable relationships. presents the framework 78 developed to represent FSCs constraints' essential interactions. 79



Figure 3. CLD (Framework) with variable relationships.

Given the context of the COVID-19 pandemic, the CLD captures in the top right of Figure Figure 3 a simple SIR (Susceptible, Infected, Recovered) model. The SIR is widely employed to model epidemics, such as Ebola, MERS, SARS, and Malaria [74]. The reinforcing loop R1 captures the contagion process that diffuses the epidemic. As the disease progresses, it infects healthy individuals closing the balancing loop B1 (depletion). 86

Among the infected individuals, some recover (e.g., balancing loop B2 – recovery) and some people die (e.g., balancing loop B3 – death). The larger the infected and dead populations, the larger the impact on economic active individuals, and the smaller the number of active workers. As the number of infected, deceased and hospitalized increase, heightened awareness and critical public opinion pressure policy makers to implement public health policies that can curb the diffusion of the epidemic (closing a balancing loop that implements social distancing measures).

Social distancing measures, quarantines, and mobility restrictions to avoid contact 94 and reduce the infection rate prevent workers (e.g., susceptible and at-risk populations) 95 from going to work, reducing the number of available workers and increasing the number 96 of employees receive salaries. An additional impact to the number of active workers in-97 cludes the infected, hospitalized, and dead. As the number of people out of work increases 98 more households face income loss, and more families are exposed to food insecurity. Fur-99 thermore, food price rises and food shortages [75], increase food insecurity to vulnerable 100 families. Balancing loop B7 (food security impact on restrictions) captures this effect (cen-101 ter, Figure 3). Families that cannot afford to buy food without a steady stream of wages 102 must return to work even if social distancing measures may still be in effect. Balancing 103 loop B6 (food security due to financial assistance) captures the need for government to 104 implement public policies to assist the vulnerable population and prevent them from fac-105 ing unprecedented problems (bottom center, Figure Figure 3). 106

As quarantines and social distancing measures reduce overall consumption and economic activity, firms face increased financial stress. Firms that have no means to retain 108 employees must lay them off (reinforcing loop R3 – layoffs due to financial stress). As 109 more firms struggle and layoffs increase, government awareness of these economic impacts also increase, leading policy-makers to provide financial assistance to enterprises 111 (loop B5 – pressure for financial assistance; at bottom center, Figure Figure 3). 112

Finally, unavailable workers and firms' stress also affects, the food supply chain 113 (FSC) productivity, captured by reinforcing loop R4 (HR & Financial Constraints; center 114 left, Figure Figure 3). Thus, producer inventories will likely face a series of disruptions 115 mentioned in the literature, encompassing disruptions in production, transportation, distribution, storage, among others (captured by loops B8 and B9; left, Figure Figure 3). 117

All variables presented in the CLD come directly from the SLR. In addition to the variables directly related to the SLR findings (e.g., HR & Financial Constraints), the CLD also captures variables (e.g., social distancing measures included in the health and safety protocols category) identified in the broader epidemiological and public policy literatures in a systemic and consistent way.

4. Discussions, implications, and research agenda

This section presents the research discussions and implications (Subsection 4.1) and the research agenda (Subsection 4.2). 125

4.1 Discussions and implications

The study highlights the impacts of pandemics and epidemics (e.g., COVID-19 pandemic) on FSCs and the policies needed to mitigate their consequences. The results reinforce the importance of identifying the different sources of impacts and implementing effective policies to ensure business continuity. The implementation of mitigation policies must be broad enough to guarantee the continuity of supply, accessibility, and distribution of essential foods. In addition, the food system (e.g., procurement, production, distribution, marketing) must be flexible enough to meet the shifting population demand, as

126

123

87

88

89

90

91

92

well as stable and continuous enough to guarantee continuous supply, while minimizing 134 the impact of possible disruptions. 135

Considering that a SC is a complex system involving different stakeholders, inter-136 ruptions in any SC link may affect all links in the chain. These impacts can be classified 137 into supply-side, demand-side, logistics and infrastructure, and management and opera-138 tion. 139

From a supply-side perspective, human resources constraints stand out as the main 140 impact caused by epidemics and pandemics. Activities automation is frequently men-141 tioned (e.g., [8,38,76]) as a policy to mitigate this impact. Food production activities must 142 be mechanized and automated using intelligent technologies so that human interference 143 is minimal. Smart technologies can be implemented to minimize labor shortages on the 144 supply chain [8]. 145

Concerning the demand-side, the most significant impacts include changes in con-146 sumer behavior, unemployment, and reduced income. Uncertainty during disasters 147 causes consumers to restrict spending, buying only necessary items. Therefore, measures 148 and indicators to monitor the demand need to be analyzed to allow a holistic view of the 149 disaster effects [47] and understanding customer needs that can be identified through dig-150 ital and analytical techniques [61]. 151

Regarding logistics and infrastructure, a critical impact deals with closures of units 152 in the supply chain. To meet specific security measures and movement restrictions, ware-153 houses, manufacturing units, wholesale markets, and other units have been forced to close 154 for different durations. The pandemic has also created additional demand for internet-155 based activities, such as online shopping, distance education, online doctors, services, and 156 home office. Therefore, measures that facilitate the work adaptation to those job opportu-157 nities are necessary for the new reality [77]. 158

From management and operations perspectives, SC instability is highlighted as the 159 top impact (e.g., fluctuations in demand and supply). In the COVID-19 pandemic, specif-160 ically, as governments intensified social distancing policies, many consumers resorted to 161 panic buying due to restrictions on movement and fear of shortages [26]. This also resulted 162 in instability in the food supply. Implementing techniques and technologies to analyze 163 supply and demand behavior is necessary to assist SC activities [61]. 164

These findings contribute to the discussion of food security and improvement in the 165 management of organizations during pandemics and epidemics. Regarding theoretical 166 implications, this study advances the discussion of impact management through a 167 broader approach, involving SC links, impact categories, and policy categories related to 168 FSCs. It lists pandemic impacts in detail and simultaneously shows respective mitigation 169 policies. Complementing the results, we developed a causal loop diagram to identify 170 cause and effect relationships between main system variables during pandemics, such as 171 the COVID-19 pandemic. In terms of practical implications, identifying potential impacts 172 of supply chain disruptions is a challenge for organizations. This paper presents both a 173 taxonomy and a framework together with a detailed list of categories that should be con-174 sidered when mapping impacts and implementing policies. Consequently, this study 175 helps create supply chains that are more resilient and prepared for possible future disrup-176 tions. 177

4.2 Research agenda

Informed by the main insights available in the SLR and grey literature, our analysis 179 identified potential gaps and highlighted several opportunities for future research in the 180 context of disaster impact assessment in FSCs: 181

- Analysis of other types of disasters; 182 183
- Analysis of vulnerable groups;
- Analysis of constraints;
- Analysis of disaster risk management in companies; 185
- Analysis of cascading effects;

178

184

•	Analysis of risk sources;	187
•	Analysis of prevention and mitigation strategies.	188

Our study focuses on mapping and understanding impacts caused by pandemics and 189 epidemics on the FSCs. However, the analysis of effects of other disasters (e.g., floods, 190 earthquakes, refugee crises, fires) allows identifying strategies to guarantee food supply 191 and minimize adverse interruption effects. In cases of earthquakes and floods, roads are 192 obstructed and transportation significantly affected. In cases of refugee crises, there is a 193 need to purchase and distribute food in a way that meets everyone's needs. Besides, some 194 social groups are more affected during disasters than others due to their high degree of 195 vulnerability (e.g., food insecure population). Identifying and analyzing vulnerability 196 groups to disasters is an essential mechanism for the assessment and implementation of 197 vulnerability mitigation policies. Regarding the FSC, plans and policies must be related to 198 ensuring food security to maintain availability, accessibility, and quality of food. 199

According to SLR results, some impacts stand out: workers' constraints, transporta-200 tion disruptions, and price instability. Specific analysis and understanding of how these 201 variables are affected can assist in decision-making and the definition of mitigation strat-202 egies. Therefore, studies addressing these topics (e.g., empirical studies) are essential tools 203 for FSC to become more resilient. Also, studies that address supply chain risk manage-204 ment (SCRM) practices in companies are crucial sources of data and information. SCRM 205 tools and practices in FSC can help business continuity and ensure food quality. SCRM is 206 also important to ensure population food safety. 207

It is essential to highlight that the entire supply chain is affected due to cascading 208 effects that occur between different echelons of the food system. Problems arising from 209 disruptions spread among suppliers, producers, retailers, wholesalers, and all links in the 210 supply chain. Thus, studies that indicate strategies and policies for flexibility and collab-211 oration among stakeholders are essential for organizations to improve information shar-212 ing and communication to improve decision making. Collaboratively, FSC links can ob-213 tain better results than can be achieved independently and can have benefits such as re-214 duced costs. 215

Besides, there are significant effects on supply chains which are due to sudden-onset 216 natural disasters (e.g., earthquakes, floods, hurricanes). However, other sources of risk 217 can cause disruptions to the flow of materials. Identifying and analyzing different sources 218 of risk, such as man-made disasters, can shed more light on mitigation measures to FSC. 219

Finally, organizations must be adequately prepared to meet changes in markets, consumer behavior, and supply chain flow caused by a disaster. For this reason, studies that assess how organizations are prepared or are preparing to respond to an emergency are necessary and provide feedback for other organizations. The mitigation plans will help them prepare for the next disruptions (e.g., pandemics and epidemics). 220

5. Conclusions

The research presents findings that can be considered a reference for stakeholders 226 involved in the pandemics and epidemics response scenario. The results concern identifying the most academically cited impacts and the proposed policies and strategies to mitigate FSCs disruptions. Also, the literature review allows the development of a CLD with 229 the identification of the main variables and their relationship, as well as a research agenda 230 with main future opportunities. 231

In analysing this research area, our study provides insights and contributions for ac-232 ademics and professionals to explore the impacts of pandemics and epidemics on food 233 chains and how the implementation of appropriate policies can mitigate the effects of a 234 disaster. The identification of SC links through a taxonomy (supply-side, demand-side, 235 logistic and infrastructure, and management and operation) facilitates understanding the 236 implementation of policies in the proper context. Consequently, it is an opportunity to 237 develop more resilient supply chains which are prepared for a certainly challenging fu-238 ture. 239

In addition to the research agenda's main topics, updating the literature review and 240 using simulation tools, such as the SD model, is also suggested. The ability to simulate the 241 dynamic behavior and to test the consequence of different policy options allows policy-242 makers to evaluate their decisions' short- and long-term results [78]. In this way, it will be 243 possible to simulate different scenarios, identified through the policies and strategies, and 244 propose appropriate and concise guidelines to help decision-makers. 245

Author Contributions: Conceptualization, F.P. and A.L; literature analysis, B.C. and L.C.; CLD246methodology, P.G and H.Y.; writing—original draft preparation, B.C and L.C; writing—review and247editing, B.C., L.C., A.L., H.Y., I.B.J., P.G., A.L. and F.P.; project administration, A.L. and F.P.; funding248acquisition, F.P. All authors have read and agreed to the published version of the manuscript.249

Funding: This research was funded by The World Bank [03.641.550/0001-88]; National Council for250Scientific and Technological Development (CNPq) [308084/2019-5, 313687/2019-6].251

Acknowledgments: The authors acknowledge the support of Coordination for the Improvement of252Higher Education Personnel (CAPES) [88887.373163/2019-00; 88887.492890/2020-00 – Finance Code253001].254

255

256

284

Conflicts of Interest: The authors declare no conflict of interest.

References

1.	Singh, S., Kumar, R., Panchal, R., & Tiwari, M. K. Impact of COVID-19 on logistics systems and disruptions in food supply	257
	chain. International Journal of Production Research, 2020, 1-16. [CrossRef]	258
2.	Dash, S. R., Mishra, U. S., & Mishra, P. Emerging issues and opportunities in disaster response supply chain management.	259
	International Journal of Supply Chain Management, 2013 , 2(1), 55-61. [CrossRef]	260
3.	Schätter, F., Hansen, O., Wiens, M., & Schultmann, F. A decision support methodology for a disaster-caused business con-	261
	tinuity management. Decision Support Systems, 2019, 118, 10-20. [CrossRef]	262
4.	Erokhin, V., & Gao, T. Impacts of COVID-19 on trade and economic aspects of food security: Evidence from 45 developing	263
	countries. International journal of environmental research and public health, 2020, 17(16), 5775. [CrossRef]	264
5.	Siche, R. What is the impact of COVID-19 disease on agriculture? Scientia Agropecuaria, 2020, 11(1), 3-6. [CrossRef]	265
6.	Sterman, J. D. Business dynamics: systems thinking and modeling for a complex world. 5th ed. McGraw-Hill, 2000, New	266
	York.	267
7.	Song, S., Goh, J. C., & Tan, H. T. Is food security an illusion for cities? A system dynamics approach to assess disturbance in	268
	the urban food supply chain during pandemics. Agricultural Systems, 2021, 189, 103045. [CrossRef]	269
8.	Chitrakar, B., Zhang, M., & Bhandari, B. Improvement strategies of food supply chain through novel food processing tech-	270
	nologies during COVID-19 pandemic, Food Control, 2021. [CrossRef]	271
9.	Fisher, Marshall. What Is the Right Supply Chain for Your Product? Harvard Business Review, 75, 105-117, 1997. [CrossRef]	272
10.	Fan, S. Agriculture, food and nutrition security under Covid-19: lessons from China. Review of Agrarian Studies, 2020,	273
	10(2369-2020-1857).	274
11.	Borsellino, V.; Kaliji, S.A.; Schimmenti, E. COVID-19 Drives Consumer Behaviour and Agro-Food Markets towards Health-	275
	ier and More Sustainable Patterns. Sustainability, 2020 12, 8366. [CrossRef]	276
12.	Montenegro, L. D., & Young, M. N. Operational Challenges in the Food Industry and Supply Chain during the COVID-19	277
	Pandemic: A Literature Review. In 2020 7th International Conference on Frontiers of Industrial Engineering (ICFIE) (pp. 1-5).	278
	IEEE, 2020. [<u>CrossRef</u>]	279
13.	Rejeb, A., Rejeb, K., & Keogh, J. G. COVID-19 and the food chain? Impacts and future research trends. <i>LogForum</i> , 2020 , 16(4).	280
14.	Kumar, A., Mangla, S. K., Kumar, P., & Song, M. Mitigate risks in perishable food supply chains: Learning from COVID-19.	281
	Technological Forecasting and Social Change, 2021a , 166, 120643. [<u>CrossRef</u>]	282
15.	Huff, A. G., Beyeler, W. E., Kelley, N. S., & McNitt, J. A. How resilient is the United States' food system to pandemics?	283

Journal of Environmental Studies and Sciences, **2015** 5(3), 337-347. [<u>CrossRef</u>]

16.	Silva, M. E., Pereira, S. C., & Gold, S. The response of the Brazilian cashew nut supply chain to natural disasters: A practice-	285
	based view. Journal of Cleaner Production, 2018, 204, 660-671. [CrossRef]	286
17.	Torraco, R. J. Writing integrative literature reviews: Using the past and present to explore the future. Human Resource De-	287
	velopment Review, 2016 , 15: 404-428. [CrossRef]	288
18.	Thomé, A. M. T.; Scavarda, L. F.; Scavarda, A. J. Conducting systematic literature review in operations management. Pro-	289
	duction Planning & Control, 2016 , 27(5): 408-420. [CrossRef]	290
19.	Mongeon, Philippe and Adèle Paul-Hus. The Journal Coverage of Web of Science and Scopus: A Comparative Analysis.	291
	Scientometrics, 2016 , 106(1):213–28. [CrossRef]	292
20.	Cooper, H. Research synthesis and meta-analysis: A step-by-step approach. Applied Social Research Methods Series 2, 2020,	293
	USA: Sage Publications, Thousand Oaks.	294
21.	Moher, D., Liberati, A., Tetzlaff, J., Altman, D.G., Prisma Group. Preferred reporting Items for systematic reviews and meta-	295
	analyses: the PRISMA statement. PLoS Med. 2009, 6 (7). [CrossRef]	296
22.	Webster, J, and Watson, R. T. Analyzing the past to prepare for the future: Writing a literature review. <i>MIS quarterly</i> , 2002,	297
	p. xiii-xxiii.	298
23.	Vom Brocke, J., A. Simons, B. Niehaves, K. Riemer, R. Plattfaut, and A. Cleven. Reconstructing the Giant: On the Importance	299
	of Rigour in Documenting the Literature Search Process. 17th European Conference on Information Systems, Verona, 2009, June	300
	8–10.	301
24.	Krippendorff, K. Content analysis: An introduction to its methodology. Sage publications 2018.	302
25.	Chodur, G. M., Zhao, X., Biehl, E., Mitrani-Reiser, J., & Neff, R. Assessing food system vulnerabilities: a fault tree modeling	303
	approach. BMC public health, 2018 , 18(1), 1-11. [CrossRef]	304
26.	Hobbs, J. E. Food supply chains during the COVID-19 pandemic. Canadian Journal of Agricultural Economics/Revue canadienne	305
	d'agroeconomie, 2020. [CrossRef]	306
27.	Sharma, R., Shishodia, A., Kamble, S., Gunasekaran, A., & Belhadi, A. Agriculture supply chain risks and COVID-19: miti-	307
	gation strategies and implications for the practitioners. <i>International Journal of Logistics Research and Applications</i> , 2020a , 1-27.	308
	[<u>CrossRef</u>]	309
28.	BBC News World (2021a). Meat factories warn Covid absences could hit supplies Available online:	310
	https://www.bbc.com/news/business-55578974 https://www.bbc.co.uk/news(accessed on 19 May 2021).	311
29.	Reuters (2020b). Explainer: How the coronavirus crisis is affecting food supply. Available online: https://www.reu-	312
	ters.com/article/us-health-coronavirus-food-explainer-idUSKBN21L0D2 (accessed on 29 November 2020).	313
30.	Reuters (2020v). UPDATE 1-Cambodia to ban some rice exports due to coronavirus Available online: https://www.reu-	314
	ters.com/article/health-coronavirus-cambodia-rice-idUSL4N2BN3EK (accessed on 29 November 2020).	315
31.	Reuters (2020e). Rising food costs lift U.S. consumer prices; coronavirus to weigh on inflation. Available online:	316
	https://www.reuters.com/article/us-usa-economy-inflation-idUSKBN20Y1U7 (accessed on 29 November 2020).	317
32.	Çakır, M., Li, Q., & Yang, X. COVID-19 and Fresh Produce Markets in the United States and China. Applied Economic Per-	318
	spectives and Policy, 2020 DOI: 10.1002/aepp.13136. [CrossRef]	319
33.	Ramakumar, R. Agriculture and the Covid-19 Pandemic: An Analysis with special reference to India. Review of Agrarian	320
	<i>Studies</i> , 2020 , 10(2369-2020-1856).	321
34.	Elleby, C., Domínguez, I. P., Adenauer, M., & Genovese, G. Impacts of the COVID-19 pandemic on the global agricultural	322
	markets. Environmental and Resource Economics, 2020, 76(4), 1067-1079. [CrossRef]	323
35.	Briano, E., Caballini, C., Giribone, P., & Revetria, R. Resiliency and vulnerability in short life cycle products' supply chains:	324
	a system dynamics model. WSEAS Transactions on Systems, 2010, 9(4), 327-337.	325
36.	Nagurney, A., Yu, M., & Qiang, Q. Supply chain network design for critical needs with outsourcing. Papers in Regional	326
	Science, 2011 , 90(1), 123-142. [CrossRef]	327

37.	Hansen, O., Friedrich, H., & Transchel, S. An inventory management approximation for estimating aggregated regional	328
	food stock levels. International Journal of Production Research, 2019, 1-17. [CrossRef]	329
38.	Sarà, G., Mangano, M. C., Berlino, M., Corbari, L., Lucchese, M., Milisenda, G., & Helmuth, B. The synergistic impacts of	330
	anthropogenic stressors and COVID-19 on aquaculture: A current global perspective. Reviews in Fisheries Science & Aquacul-	331
	<i>ture</i> , 2021 , 1-13. [<u>CrossRef</u>]	332
39.	Höhler, J., & Lansink, A. O. Measuring in the impact of COVID-19 on stock prices and profits in the food supply chain.	333
	Agribusiness, 2020. [CrossRef]	334
40.	Qingbin, Wang., Liu, C. Q., Zhao, Y. F., Kitsos, A., Cannella, M., Wang, S. K., & Lei, H. A. N. Impacts of the COVID-19	335
	pandemic on the dairy industry: Lessons from China and the United States and policy implications. Journal of Integrative	336
	<i>Agriculture</i> , 2020 , 19(12), 2903-2915. [<u>CrossRef</u>]	337
41.	Farcas, A. C., Galanakis, C. M., Socaciu, C., Pop, O. L., Tibulca, D., Paucean, A., & Socaci, S. A. Food Security during the	338
	Pandemic and the Importance of the Bioeconomy in the New Era. Sustainability, 2021, 13(1), 150. [CrossRef]	339
42.	Chari, F., & Ngcamu, B. S. An assessment of the impact of disaster risks on dairy supply chain performance in Zimbabwe.	340
	<i>Cogent Engineering</i> , 2017 , 4(1), 1409389. [<u>CrossRef</u>]	341
43.	Barcaccia, G., D'Agostino, V., Zotti, A., & Cozzi, B. Impact of the SARS-CoV-2 on the Italian Agri-Food Sector: An Analysis	342
	of the Quarter of Pandemic Lockdown and Clues for a Socio-Economic and Territorial Restart. Sustainability, 2020, 12(14),	343
	5651. [<u>CrossRef]</u>	344
44.	Chowdhury, M. T., Sarkar, A., Paul, S. K., & Moktadir, M. A. A case study on strategies to deal with the impacts of COVID-	345
	19 pandemic in the food and beverage industry. Operations Management Research, 2020, 1-13. [CrossRef]	346
45.	Reuters (2020ao). Nestle CEO says pandemic has made management more hands-on. Available online: https://www.reu-	347
	ters.com/article/uk-nestle-ceo-idUKKBN26K28T (accessed on 30 November 2020).	348
46.	Nasution, A., Azmi, N. N., Ananda, F. P., & Azalia, M. Disturbance Management Strategy in the Food Supply Chain in The	349
	Middle of Pandemic COVID-19. In IOP Conference Series: Materials Science and Engineering, 2020, 1003(1): 012133). [CrossRef]	350
47.	Knorr, D., & Khoo, C. S. H. COVID-19 and Food: Challenges and Research Needs. Frontiers in Nutrition, 2020, 7. [CrossRef]	351
48.	Biswas, T. K., & Das, M. C. Selection of the barriers of supply chain management in Indian manufacturing sectors due to	352
	COVID-19 impacts. Operational Research in Engineering Sciences: Theory and Applications, 2020, 3(3), 1-12. [CrossRef]	353
49.	Kansiime, M. K., Tambo, J. A., Mugambi, I., Bundi, M., Kara, A., & Owuor, C. COVID-19 implications on household income	354
	and food security in Kenya and Uganda: Findings from a rapid assessment. World development, 2021, 137, 105199. [CrossRef]	355
50.	Dombroski, K., Diprose, G., Sharp, E., Graham, R., Lee, L., Scobie, M., & Martin-Neuninger, R. Food for People in Place:	356
	Reimagining Resilient Food Systems for Economic Recovery. Sustainability, 2020, 12(22), 9369. [CrossRef]	357
51.	Kumar, S. Planning for avian flu disruptions on global operations: a DMAIC case study. International journal of health care	358
	<i>quality assurance</i> , 2012 . [CrossRef]	359
52.	Gray, R. S. Agriculture, transportation, and the COVID-19 crisis. Canadian Journal of Agricultural Economics/Revue canadi-	360
	enne d'agroeconomie, 2020 , 68(2), 239-243. [<u>CrossRef</u>]	361
53.	Pulighe, G., & Lupia, F. Food first: COVID-19 outbreak and cities lockdown a booster for a wider vision on urban agriculture.	362
	<i>Sustainability</i> , 2020 , 12(12), 5012. [<u>CrossRef</u>]	363
54.	Richards, T. J., & Rickard, B. COVID-19 impact on fruit and vegetable markets. Canadian Journal of Agricultural Economics/Re-	364
	vue canadienne d'agroeconomie, 2020. [<u>CrossRef</u>]	365
55.	Nordhagen, S., Igbeka, U., Rowlands, H., Shine, R. S., Heneghan, E., & Tench, J. COVID-19 and small enterprises in the food	366
	supply chain: Early impacts and implications for longer-term food system resilience in low-and middle-income countries.	367
	World Development, 2021 , 141, 105405. [<u>CrossRef</u>]	368

56.	Janssen, M., Chang, B. P., Hristov, H., Pravst, I., Profeta, A., & Millard, J. Changes in Food Consumption During the COVID-	369
	19 Pandemic: Analysis of Consumer Survey Data From the First Lockdown Period in Denmark, Germany, and Slovenia.	370
	Frontiers in nutrition, 2021 , 8: 60. [CrossRef]	371
57.	Clapp, J., & Moseley, W. G. This food crisis is different: COVID-19 and the fragility of the neoliberal food security order. <i>The</i>	372
	Journal of Peasant Studies, 2020, 47(7), 1393-1417. [<u>CrossRef]</u>	373
58.	Surni, Nendissa, D. R., Wahib, M. A., Astuti, M. H., Arimbawa, P., Kapa, M. M., & Elbaar, E. F. Socio-economic impact of	374
	the Covid-19 pandemic: Empirical study on the supply of chicken meat in Indonesia. AIMS Agriculture and Food, 2020, 6(1),	375
	65-81. [<u>CrossRef]</u>	376
59.	CNN (2020a). Americans should "only purchase enough food and essentials for the week ahead," FDA says. Available	377
	online: https://edition.cnn.com/world/live-news/coronavirus-outbreak-03-17-20-intl-	378
	hnk/h_786a3e672eead3d36584d8321b124ca0 (accessed on 19 May 2021).	379
60.	Fan, S., Teng, P., Chew, P., Smith, G., & Copeland, L. Food system resilience and COVID-19–Lessons from the Asian expe-	380
	rience. Global Food Security, 2021, 28, 100501. [CrossRef]	381
61.	Telukdarie, A., Munsamy, M., & Mohlala, P. Analysis of the Impact of COVID-19 on the Food and Beverages Manufacturing	382
	Sector. Sustainability, 2020, 12(22), 9331. [CrossRef]	383
62.	Schätter, F., Hansen, O., Wiens, M., & Schultmann, F. A decision support methodology for a disaster-caused business con-	384
	tinuity management. Decision Support Systems, 2019, 118, 10-20. [CrossRef]	385
63.	Magar, D. B. T., Pun, S., Pandit, R., & Rola-Rubzen, M. F. Pathways for building resilience to COVID-19 pandemic and	386
	revitalizing the Nepalese agriculture sector. Agricultural Systems, 2021, 187, 103022. [CrossRef]	387
64.	Nchanji, E. B., Lutomia, C. K., Chirwa, R., Templer, N., Rubyogo, J. C., & Onyango, P. Immediate impacts of COVID-19	388
	pandemic on bean value chain in selected countries in sub-Saharan Africa. Agricultural Systems, 2021, 188, 103034. [CrossRef]	389
65.	Grinberga-Zalite, G., Pilvere, I., Muska, A., & Kruzmetra, Z. Resilience of Meat Supply Chains during and after COVID-19	390
	Crisis. Emerging Science Journal, 2021 , 5(1), 57-66. [CrossRef]	391
66.	Coluccia, B., Agnusdei, G. P., Miglietta, P. P., & De Leo, F. Effects of COVID-19 on the Italian agri-food supply and value	392
	chains. Food Control, 2021, 107839. [CrossRef]	393
67.	Deaton, B. J., & Deaton, B. J. Food security and Canada's agricultural system challenged by COVID-19. Canadian Journal of	394
	Agricultural Economics/Revue canadienne d'agroeconomie, 2020 , 68(2), 143-149.	395
68.	Márquez, R., Tolosa, L., & Celis, M. T. Understanding COVID-19 effect on the US supply chain of strategic products: im-	396
	portant factors, current situation, and fu-ture perspective Entendiendo el efecto de COVID-19 en las cadenas de suministro	397
	de productos estratégicos en EE. UU.: factores clave, situación. Revista Ciencia e Ingeniería. 2021, 42(1).	398
69.	Boyacı-Gündüz, C. P., Ibrahim, S. A., Wei, O. C., & Galanakis, C. M. Transformation of the Food Sector: Security and Resil-	399
	ience during the COVID-19 Pandemic. Foods, 2021, 10(3), 497. [CrossRef]	400
70.	Yoshizaki, H.T.Y.; de Brito Junior, I.; Hino, C.M.; Aguiar, L.L.; Pinheiro, M.C.R. Relationship between Panic Buying and Per	401
	Capita Income during COVID-19. Sustainability 2020, 12, 9968 [CrossRef]	402
71.	Ihle, R., Rubin, O. D., Bar-Nahum, Z., & Jongeneel, R. Imperfect food markets in times of crisis: economic consequences of	403
	supply chain disruptions and fragmentation for local market power and urban vulnerability. Food Security, 2020, 12(4), 727-	404
	734. [<u>CrossRef]</u>	405
72.	Udmale, P., Pal, I., Szabo, S., Pramanik, M., & Large, A. Global food security in the context of COVID-19: A scenario-based	406
	exploratory analysis. Progress in Disaster Science, 2020, 7, 100120. [CrossRef]	407
73.	Rozaki, Z. COVID-19, Agriculture, and Food Security in Indonesia. Reviews in Agricultural Science, 2020, 8, 243-260. [Cross-	408
	<u>Ref</u>	409
74.	Sytem Dynamics Society – SDS, 2020. COVID-19 Resource Page. Available at: https://www.systemdynamics.org/covid-19	410
	(access July, 18, 2020).	411

75.	Akseer, N., Kandru, G., Keats, E. C., & Bhutta, Z. A. COVID-19 pandemic and mitigation strategies: implications for mater-	412
	nal and child health and nutrition. The American journal of clinical nutrition, 2020. [CrossRef]	413
76.	Nchanji, E. B., & Lutomia, C. K. Regional impact of COVID-19 on the production and food security of common bean small-	414
	holder farmers in Sub-Saharan Africa: Implication for SDG's. Global Food Security, 2021, 29, 100524. [CrossRef]	415
77.	Rasul, G., Nepal, A. K., Hussain, A., Maharjan, A., Joshi, S., Lama, A., & Sharma, E. Socio-Economic Implications of	416
	COVID-19 Pandemic in South Asia: Emerging Risks and Growing Challenges. Front. Sociol, 2021, 6: 629693. [CrossRef]	417
78.	Gonçalves, P. Balancing provision of relief and recovery with capacity building in humanitarian operations. Operations Man-	418
	agement Research, 2011 , 4(1-2): 39-50. [<u>CrossRef]</u>	419
79.	Liverpool-Tasie, L. S. O., Reardon, T., & Belton, B. Essential non-essentials: COVID-19 policy missteps in N igeria rooted in	420
	persistent myths about A frican food supply chains. Applied Economic Perspectives and Policy, 2021, 43(1), 205-224. [CrossRef]	421
80.	Hashem, N. M., González-Bulnes, A., & Rodriguez-Morales, A. J. Animal welfare and livestock supply chain sustainability	422
	under the COVID-19 outbreak: An overview. Frontiers in veterinary science, 2020, 7, 679. [CrossRef]	423
81.	Min, S., Zhang, X., & Li, G. A snapshot of food supply chain in Wuhan under the COVID-19 pandemic. China Agricultural	424
	Economic Review, 2020.	425
82.	Weersink, A., von Massow, M., Bannon, N., Ifft, J., Maples, J., McEwan, K., & Wood, K. COVID-19 and the agri-food	426
	system in the United States and Canada. Agricultural Systems, 2021, 188, 103039. [CrossRef]	427
83.	Sukhwani, V., Deshkar, S., & Shaw, R. Covid-19 lockdown, food systems and urban-rural partnership: Case of Nagpur,	428
	India. International journal of environmental research and public health, 2020, 17(16), 5710. [CrossRef]	429
84.	Butu, A., Brumă, I. S., Tanasă, L., Rodino, S., Dinu Vasiliu, C., Doboș, S., & Butu, M. The impact of COVID-19 crisis upon	430
	the consumer buying behavior of fresh vegetables directly from local producers. Case study: the quarantined area of suceava	431
	county, Romania. International journal of environmental research and public health, 2020, 17(15), 5485. [CrossRef]	432
85.	Zhang, Y., Diao, X., Chen, K. Z., Robinson, S., & Fan, S. Impact of COVID-19 on China's macroeconomy and agri-food	433
	system-an economy-wide multiplier model analysis. China Agricultural Economic Review, 2020. [CrossRef]	434
86.	Xu, Z., Elomri, A., El Omri, A., Kerbache, L., & Liu, H. The Compounded Effects of COVID-19 Pandemic and Desert Locust	435
	Outbreak on Food Security and Food Supply Chain. Sustainability, 2021, 13(3), 1063. [CrossRef]	436
87.	Xu, Z., Elomri, A., Kerbache, L., & El Omri, A. (2020). Impacts of COVID-19 on global supply chains: facts and perspectives.	437
	IEEE Engineering Management Review, 2020, 48(3), 153-166.	438
88.	Jamwal, A., & Phulia, V. Multisectoral one health approach to make aquaculture and fisheries resilient to a future pandemic-	439
	like situation. <i>Fish and Fisheries</i> , 2021 , 22(2), 449-463. [CrossRef]	440
89.	Martinez, C. C., Maples, J. G., & Benavidez, J. Beef Cattle Markets and COVID-19. Applied Economic Perspectives and Policy,	441
	2021 , 43(1), 304-314. [<u>CrossRef</u>]	442
90.	Love, D. C., Allison, E. H., Asche, F., Belton, B., Cottrell, R. S., Froehlich, H. E., & Zhang, W. (2021). Emerging COVID-19	443
	impacts, responses, and lessons for building resilience in the seafood system. Global Food Security, 2021. [CrossRef]	444
91.	Nakat, Z., & Bou-Mitri, C. COVID-19 and the food industry: Readiness assessment. Food control, 2021, 107661. [CrossRef]	445
92.	Fan, S., Si, W., & Zhang, Y. How to prevent a global food and nutrition security crisis under COVID-19? China Agricultural	446
	Economic Review, 2020. [CrossRef]	447
93.	Brewin, D. G. The impact of COVID-19 on the grains and oilseeds sector. Canadian Journal of Agricultural Economics/Revue	448
	canadienne d'agroeconomie, 2020 , 68(2), 185-188. [<u>CrossRef]</u>	449
94.	Nandi, R., Nedumaran, S., Selvaraj, A., Datta Mazumdar, S., & Kumar, S. The COVID-19 Induced Disruptions across	450
	Groundnut Value Chain: Empirical Evidence from South India. Sustainability, 2021, 13(4), 1707. [CrossRef]	451
95.	Mishra, A., Bruno, E., & Zilberman, D. Compound natural and human disasters: Managing drought and COVID-19 to sus-	452
	tain global agriculture and food sectors. Science of the Total Environment, 2021, 754, 142210. [CrossRef]	453

96. Whelan, J., Brown, A. D., Coller, L., Strugnell, C., Allender, S., Alston, L., & Bell, C. The Impact of COVID-19	on Rural 454	
Food Supply and Demand in Australia: Utilising Group Model Building to Identify Retailer and Customer Per	spectives. 455	
Nutrients, 2021 , 13(2), 417. [<u>CrossRef]</u>	456	
97. Kumar, P., Singh, S. S., Pandey, A. K., Singh, R. K., Srivastava, P. K., Kumar, M., & Drews, M. Multi-level impa	acts of the 457	
COVID-19 lockdown on agricultural systems in India: The case of Uttar Pradesh. Agricultural Systems, 2021b, 18	7, 103027. 458	
[<u>CrossRef]</u>	459	
98. Falkendal, T., Otto, C., Schewe, J., Jägermeyr, J., Konar, M., Kummu, M., & Puma, M. J. Grain export restriction	ns during 460	
COVID-19 risk food insecurity in many low-and middle-income countries. Nature Food, 2021, 2(1), 11-14. [CrossRe	<u>ef]</u> 461	
99. Mahajan, K., & Tomar, S. COVID-19 and Supply Chain Disruption: Evidence from Food Markets in India. Americ	an journal 462	
of agricultural economics, 2021, 103(1), 35-52. [CrossRef]	463	
100. Khan, S. A. R., Razzaq, A., Yu, Z., Shah, A., Sharif, A., & Janjua, L. Disruption in food supply chain and undernou	arishment 464	
challenges: An empirical study in the context of Asian countries. Socio-Economic Planning Sciences, 2021, 101033. [<u>CrossRef</u>] 465	
101. Matthews, A. EU Food System Strengths and Vulnerabilities during Covid-19. EuroChoices, 2020, 19(3), 4-12. [Cross	<u>ssRef]</u> 466	
102. Swinnen, J., & McDermott, J. Covid-19 and Global Food Security. EuroChoices, 2020. [CrossRef]	467	
103. Lin, B. X., & Zhang, Y. Y. Impact of the COVID-19 pandemic on agricultural exports. Journal of Integrative Agricult	ture, 2020, 468	
19(12), 2937-2945. [<u>CrossRef]</u>	469	
104. Deconinck, K., Avery, E., & Jackson, L. A. Food Supply Chains and Covid-19: Impacts and Policy Lessons. EuroCho	nices, 2020, 470	
19(3), 34-39. [<u>CrossRef]</u>	471	
105. Rowan, N. J., & Galanakis, C. M. Unlocking challenges and opportunities presented by COVID-19 pandemic	for cross- 472	
cutting disruption in agri-food and green deal innovations: Quo Vadis? Science of the Total Environment, 2020, 14136	52. [<u>Cross-</u> 473	
<u>Ref</u>]	474	
106. Narayanan, S., & Saha, S. Urban food markets and the COVID-19 lockdown in India. Global Food Security, 2021, 2	.9, 100515. 475	
[CrossRef]	476	
107. Aday, S., & Aday, M. S. (2020). Impact of COVID-19 on the food supply chain. Food Quality and Safety, 2020, 4(4)), 167-180. 477	
[CrossRef]	478	
108. Paul, M., Baritaux, V., Wongnarkpet, S., Poolkhet, C., Thanapongtharm, W., Roger, F., & Ducrot, C. Practices a	ssociated 479	
with highly pathogenic avian influenza spread in traditional poultry marketing chains: social and economic pers	spectives. 480	
<i>Acta tropica</i> , 2013 , 126(1), 43-53.	481	
109. Orden, D. Resilience and Vulnerabilities of the North American Food System during the Covid-19 Pandemic. Eu	roChoices, 482	
2020. [<u>CrossRef]</u>	483	
110. de Paulo Farias, D., & de Araújo, F. F. Will COVID-19 affect food supply in distribution centers of Brazilian region	is affected 484	
by the pandemic?. Trends in Food Science & Technology, 2020 ,103, 361-366. [CrossRef]	485	
111. Sers, C. F., & Mughal, M. Covid-19 outbreak and the need for rice self-sufficiency in West Africa. World Development	<i>ient</i> , 2020 , 486	
135, 105071. [<u>CrossRef]</u>	487	
112. Paganini, N., Adinata, K., Buthelezi, N., Harris, D., Lemke, S., Luis, A., & Stöber, S. Growing and Eating Food d	luring the 488	
COVID-19 Pandemic: Farmers' Perspectives on Local Food System Resilience to Shocks in Southern Africa and I	ndonesia. 489	
<i>Sustainability</i> , 2020 , 12(20), 8556. [CrossRef]	490	
113. Cao, L., Li, T., Wang, R., & Zhu, J. Impact of COVID-19 on China's agricultural trade. China Agricultural Econom	ic Review, 491	
2020.	492	
	 Wielan J.; Burden, P. D.; Collie, P.; Studgner, C.; Michael, S.; Michael, S.; Michael, Y.; Li, C. and End, C. The implact of COVID-19 Food Supply and Demand in Australia: Utilising Group Model Building to Identify Retailer and Customer Per Nutrienis, 2021, 13(2), 417. [CrossRef] Kumar, P.; Singh, S. S.; Pandey, A. K.; Singh, R. K.; Srivastava, P. K.; Kumar, M.; & Drews, M. Multi-level impr COVID-19 lockdown on agricultural systems in India: The case of Uttar Pradesh. <i>Agricultural Systems</i>, 2021b, 18 [CrossRef] Falkendal, T.; Otto, C.; Schewe, J.; Jägermeyr, J.; Konar, M.; Kummu, M.; & Puma, M. J. Grain export restrictio COVID-19 risk food insecurity in many low-and middle-income countries. <i>Nature Food</i>, 2021, 2(1), 11-14. [CrossRef] Mahajan, K., & Tomar, S. COVID-19 and Supply Chain Disruption: Evidence from Food Markets in India. <i>Americ of agricultural economics</i>, 2021, 103(1), 35-52. [CrossRef] Khan, S. A. R., Razzad, A. Yu, Z., Shah, A.; Sharif, A., & Janjua, L. Disruption in food supply chain and undernot challenges: An empirical study in the context of Asian countries. <i>Socio-Economic Planning Sciences</i>, 2021, 101033. [5 101. Matthews, A. RU Food System Strengths and Vulnerabilities during Covid-19. <i>EuroChoices</i>, 2020, [CrossRef] Lin, B. X., & Zhang, Y. Y. Impact of the COVID-19 pandemic on agricultural exports. <i>Journal of Integrative Agricult</i> 19(2), 2937-2945. [CrossRef] LocossRef] Rowan, N. J., & Galanakis, C. M. Unlocking challenges and opportunities presented by COVID-19 pandemic cutting disruption in agri-food and green deal innovations: Quo Vadis? <i>Science of the Total Environment</i>, 2020, 14136 Ref] Rovan, N. J., & Galanakis, C. M. Unlocking challenges and opportunities presented by COVID-19 pandemic cutting disruption in agri-food and green deal innovations: Quo Vadis? <i>Science of the Total Environment</i>, 2020, 14136 Ref] Roaul, M., Baritaux, V., Wongnarkpet,	 Wienard J, Bowit P, D. Collis, E., Mugaled, C., Minghell, S., Makol, E., Jie & Deli, C. Inferingtot of COVID-19 of Kalla 44 Pood Surply and Demand in Australia: Utilising Group Model Building to Identify Retailer and Customer Perspectives. Nutrients, 2021, 13(2), 417. [CrossRef] Kumar, P., Singh, S. S., Pandey, A. K., Singh, R. K., Srivastava, P. K., Kumar, M., & Drews, M. Multi-level impacts of the 457 (COVID-19 lockdown on agricultural systems in India: The case of Ultar Pradesh. <i>Agricultural Systems</i>, 2021b, 187, 103027. Falkendal, T., Oto, C., Schewe, J., Jägermeyr, J., Konar, M., Kummu, M., & Puma, M. J. Grain export restrictions during 469 (COVID-19 risk food insecurity in many low-and middle-income countries. <i>Nature Food</i>, 2021, 1014. [CrossRef] Mahajan, K., & Tomar, S. (COVID-19 and Supply Chain Disruption: Evidence from Food Markets in India. <i>American journal 462 dyticultural economics</i>, 2021, 103(1), 35-52. [CrossRef] Mathiews, A. & UF Ood System Strengths and Unherabilities during Covid-19. <i>BuroChoices</i>, 2021, 101033. [CrossRef] Mathews, A. & UF ood System Strengths and Unherabilities during Covid-19. <i>BuroChoices</i>, 2021, 101033. [CrossRef] Mathews, A. & Zhang, Y. Y. Impact of the COVID-19 pandemic on agricultural exports. <i>Journal of Integrative Agriculture</i>, 2020, 14362. [CrossRef] Swinnen, J., & McDermott, J. Covid-19 and Global Food Security. <i>EuroChoices</i>, 2020, 19(3), 41-21. [CrossRef] Acowan, N. J., & Galanakis, C. M. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-curting disruption in agri-food and green deal innovations: Quo Valis? <i>Science of the Total Environment</i>, 2020, 14(36, [CrossRef] Rovany, N. J., & Galanakis, C. M. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-curting disruption in agri-food and green deal innovations: Quo Valis? <i>Science of the Total </i>

- 114. Alam, G. M., & Khatun, M. N. Impact of COVID-19 on vegetable supply chain and food security: Empirical evidence from493Bangladesh. Plos one, 2021, 16(3), e0248120. [CrossRef]494
- 115. Lawson-Lartego, L., & Cohen, M. J. 10 recommendations for African governments to ensure food security for poor and vulnerable populations during COVID-19. *Food Security*, 2020, 12(4), 899-902. [CrossRef]
 496

116. Larue, B. Labor issues and COVID-19. Canadian Journal of Agricultural Economics/Revue canadienne d'agroeconomie, 2020,	497
68(2), 231-237. [<u>CrossRef]</u>	498
117. Abiral, B., & Atalan-Helicke, N. Trusting food supply chains during the pandemic: reflections from Turkey and the US. Food	499
and Foodways, 2020 , 28(3), 226-236. [<u>CrossRef]</u>	500
118. Nkamleu, G. B. African agriculture in the context of COVID-19: Finding salvation in the devil. African Journal of Agricultural	501
and Resource Economics, 2020.	502
119. Hayes, D. J., Schulz, L. L., Hart, C. E., & Jacobs, K. L. A descriptive analysis of the COVID-19 impacts on US pork, turkey,	503
and egg markets. Agribusiness, 2021, 37(1), 122-141. [CrossRef]	504
120. Kerr, W. A. The COVID-19 pandemic and agriculture: Short-and long-run implications for international trade relations.	505
Canadian Journal of Agricultural Economics/Revue canadienne d'agroeconomie, 2020 , 68(2), 225-229.	506
121. Pu, M., & Zhong, Y. Rising concerns over agricultural production as COVID-19 spreads: Lessons from China. Global food	507
security, 2020 , 26, 100409. [CrossRef]	508
122. Widi, A., Sari, E. D., & Jahroh, S. The Change of Fruit Supply Chain in Response to Covid-19 Pandemic in West Java, Indo-	509
nesia (Case Study of Anto Wijaya Fruit). In Journal of Physics: Conference Series. IOP Publishing, 2021 , 1764(1):012036. [Cross-	510
Ref]	511
123. Ali, M. H., Suleiman, N., Khalid, N., Tan, K. H., Tseng, M. L., & Kumar, M. Supply chain resilience reactive strategies for	512
food SMEs in coping to Covid-19 crisis. <i>Trends in Food Science & Technology</i> , 2021 . [CrossRef]	513
124. Cavallo, C., Sacchi, G., & Carfora, V. Resilience effects in food consumption behaviour at the time of Covid-19: perspectives	514
from Italy. <i>Helivon</i> , 2020 , 6(12), e05676. [CrossRef]	515
125. Ibn-Mohammed, T., Mustapha, K. B., Godsell, I. M., Adamu, Z., Babatunde, K. A., Akintade, D. D., & Koh, S. C. L. A	516
critical review of the impacts of COVID-19 on the global economy and ecosystems and opportunities for circular economy	517
strategies. Resources. Conservation and Recucling, 105169, 2020. [CrossRef]	518
126. Marchant-Forde, I. N., & Boyle, L. A. COVID-19 effects on livestock production: A One Welfare issue. <i>Frontiers in veterinary</i>	519
science. 2020 , 7, 734. [CrossRef]	520
127. Farrell, P., Thow, A. M., Wate, I. T., Nonga, N., Vatucawaga, P., Brewer, T., & Andrew, N. L. COVID-19 and Pacific food	521
system resilience: opportunities to build a robust response. <i>Food Security</i> 2020 12(4), 783-791 [CrossRef]	522
128 Lederman R. Kurnia S. & Lederman I. Designing supply chain systems to cone with catastrophes. <i>PACIS</i> 2009 Proceedings	523
2009 92	524
129 Mohan LL Viswanadham N & Trikha P Impact of Avian Influenza in the Indian poultry industry: A supply chain risk	525
perspective International Journal of Logistics Systems and Management 2009 5(1-2) 89-105	525
130 BBC News World (2020a) Coronavirus: Food supply chains 'need a rethink' Available online:	520
https://www.bbc.com/news/uk_wales_52500906https://www.bbc.co.uk/news (accessed on 02 December 2020)	528
131 BBC News World (2020b) Coronavirus: Five ways the outbreak is hitting global food industry Available online:	520
https://www.bbc.com/news/world-52267943https://www.bbc.co.uk/news/accessed on 02 December 2020)	530
132 BBC News World (2020c) Online retailer Ocado warns of shortages as suppliers cut choice. Available online:	531
https://www.bbc.com/paus/bucipess 55600215 (accessed on 02 December 2020)	531
122 BBC, Nouse World (2020d), Cavid 10: 'Sarious diaruntian' forred as Dever balts traffic to France, Available onlines	552
https://www.hb.com/neurofluciness E520505 (accessed on 02 December 2020)	533
<u>https://www.bbc.com/news/business-55589505</u> (accessed on 02 December 2020).	534
bttrac//www.bbc.com/marge/marge/marges/5502082 (accessed or 02 December 2020)	535
<u>nttps://www.bbc.com/news/business-55293982 (</u> accessed on 02 December 2020).	536
135. DDC INews world (2020). Coronavirus: will Covid-19 speed up the use of robots to replace human workers? Available	537
online: <u>nttps://www.bbc.com/news/technology-52340651</u> (accessed on 02 December 2020).	538

136. BBC News World (2020g). Coronavirus: Supermarkets 'drastically' cutting product ranges. Available online:	539
https://www.bbc.com/news/business-51961624 (accessed on 03 December 2020).	540
137. BBC News World (2020h). How dependent is the UK on the EU for food? Available online: https://www.bbc.com/news/busi-	541
<u>ness-55408788 (</u> accessed on 03 December 2020).	542
138. BBC News World (2020i). Coronavirus: Meat shortage leaves US farmers with 'mind-blowing' choice. Available online:	543
https://www.bbc.com/news/world-us-canada-52575904 (accessed on 03 December 2020).	544
139. BBC News World (2020j). Will coronavirus lockdown cause food shortages in India? Available online:	545
https://www.bbc.com/news/world-asia-india-52176564 (accessed on 03 December 2020).	546
140. BBC News World (2021b). Supermarkets try to calm food shortage fears amid border chaos. Available	547
online:https://www.bbc.co.uk/news https://www.bbc.com/news/business-55393076 (accessed on 19 May 2021).	548
141. BBC News World (2021c). Funding pledge to help seafood industry cope with Covid and Brexit. Available online:	549
https://www.bbc.com/news/uk-scotland-scotland-business-55912863 (accessed on 19 May 2021).	550
142. CNN News World (2020b). Oxfam: A hunger crisis linked to Covid-19 could kill more people than the disease itself. Avail-	551
able online: https://edition.cnn.com/world/live-news/coronavirus-pandemic-07-12-20-	552
intl/h_ae4973d507d72a09a42e29be97c7df2c (accessed on 05 December 2020).	553
143 CNN News World (2020c) UK working closely with international partners to minimize travel disruptions. Available online:	554
https://edition.com/world/live-news/coronavirus-pandemic-vaccine-updates-12-21-	555
20/h ee0b06c199f98a3e0d84ed81129b70a0 (accessed on 05 December 2020)	556
144 CNN News World (2020d). Opinion: What we at America's FDA are doing to fight Covid-19. Available online: https://edi-	557
tion cnn com/world/live-news/coronavirus-nandemic-03-31-20/h d2dbc899339547a87edfb23db80fa7ac (accessed on 19	558
May 2021)	559
145 Reuters (2020a) Coronavirus upends global food supply chains in latest economic shock. Available online: https://www.reu-	560
ters com/article/us-health-coronavirus-food-supplies-insi-idLISKBN21L2V7 (accessed on 19 May 2021)	561
146 Reuters (2020c) Western supply chains buckle as coronavirus lockdowns spread. Available online: https://www.reu-	562
ters com/article/us-bealth-coronavirus-freight-idUSKBN21A2PB (accessed on 05 December 2020)	563
147 Routors (2020d) How Africa risks realing from a health crisis to a food crisis. Available opling: https://www.rou	564
tars com/article/us health compressions food of its insich idUSVCN2260M2 (correspond on 10 May 2021)	564
148 Bentary (2020a) Dising food easts lift U.C. company or prices companying to use in flation. Assoluble onlines	565
148. Reuters (2020e). Rising food costs int U.S. consumer prices; coronavirus to weign on inflation. Available online:	566
<u>https://www.reuters.com/article/us-usa-economy-inflation-id/USKBN209107 (</u> accessed on 19 May 2021).	567
149. Reuters (2020f). Coronavirus will constrain U.S. meat supply despite Trump order: Tyson Foods. Available online:	568
https://www.reuters.com/article/us-tyson-toods-results-idUSKBN22G1D1 (accessed on 19 May 2021).	569
150. Reuters (2020g). Lebanon import woes deepen as supply chains buckle under coronavirus. Available online:	570
https://www.reuters.com/article/us-health-coronavirus-lebanon-supplies-idUSKBN21Q1XE (accessed on 19 May 2021).	571
151. Reuters (2020h). Costco limits meat purchases as supply shortages loom. Available online: <u>https://www.reuters.com/arti-</u>	572
<u>cle/health-coronavirus-costco-wholesale-idUKL4N2CM288</u> (accessed on 05 December 2020).	573
152. Reuters (2020i). Bunge reports loss, cuts 2020 outlook as coronavirus hits demand. Available: https://www.reuters.com/ar-	574
ticle/us-bunge-results/bunge-reports-loss-cuts-2020-outlook-as-coronavirus-hits-demand-idUKKBN22I1EY (accessed on	575
19 May 2021).	576
153. Reuters (2020j). U.S. dairy farmers dump milk as pandemic upends food markets. Available online: https://www.reu-	577
ters.com/article/us-health-coronavirus-dairy-insight-idUSKBN21L1DW (accessed on 17 May 2021).	578
154. Reuters (2020k). Coronavirus spreads among fruit and vegetable packers, worrying U.S. officials. Available online:	579
https://www.reuters.com/article/us-health-coronavirus-usa-farmworkers-idUSKBN23I1FO (accessed on 19 May 2021).	580

155. Reuters (20201). West African food trade under strain as COVID-19 shuts borders. Available online: https://www.reu-	581
ters.com/article/us-health-coronavirus-food-africa-idUSKBN2330RV (accessed on 19 May 2021).	582
156. Reuters (2020m). EU trade chief urges for more diverse supply chains after crisis. Available online: <u>https://www.reu-</u>	583
ters.com/article/us-health-coronavirus-eu-trade-idUSKCN21Y27P (accessed on 17 May 2021).	584
157. Reuters (2020n). Side effects: Fuel demand crash shuts U.S. ethanol plants, meatpackers lack refrigerant. Available online:	585
https://www.reuters.com/article/us-health-coronavirus-carbon-dioxide-idUSKBN21P2DK (accessed on 19 May 2021).	586
158. Reuters (2020o). Latin America to see sharp increase in hunger, poverty: FAO. Available online: https://www.reu-	587
ters.com/article/us-health-coronavirus-fao/latin-america-to-see-sharp-increase-in-hunger-poverty-fao-idUKKCN22A1WV	588
(accessed on 19 May 2021).	589
159. Reuters (2020p). Exclusive: Indian rice exports suspended on supply chain disruption – industry. Available online:	590
https://www.reuters.com/article/us-health-coronavirus-india-food-exclusi-idUSKBN21L1XX (accessed on 19 May 2021).	591
160. Reuters (2020q). Three meat plants shut in Iowa and Pennsylvania as coronavirus outbreak spreads. Available online:	592
https://www.reuters.com/article/us-health-coronavirus-usa-meat-idUSKBN21O2J5 ((accessed on 17 May 2021).	593
161. Reuters (2020r). India's truckers in crisis: Police checks, no food and fears of coronavirus. Available online:	594
https://www.reuters.com/article/us-health-coronavirus-india-freight-idUSKCN2260VE (accessed on 19 May 2021).	595
162. Reuters (2020s). How COVID-19 affected U.S. consumer prices in March. Available online: https://www.reuters.com/arti-	596
cle/us-health-coronavirus-usa-prices-idUSKCN21S20H (accessed on 19 May 2021).	597
163. Reuters (2020t). U.S. consumer prices post largest drop in five years amid coronavirus disruptions. Available online:	598
https://www.reuters.com/article/us-usa-economy-economy-idUSKCN21S1B7 (accessed on 19 May 2021).	599
164. Reuters (2020u). U.S. details up to \$14 billion in new aid for farmers. Available online: https://www.reuters.com/article/us-	600
health-coronavirus-usa-farmers-aid-idUSKBN269261 (accessed on 17 May 2021).	601
165. Reuters (2020w). Food security concerns stoked as exporters curb sales, importers buy more. Available online:	602
https://www.reuters.com/article/us-health-coronavirus-food-security-idUSKBN21D0YV (accessed on 19 May 2021).	603
166. Reuters (2020x). Trump administration to unveil \$15.5 billion first phase of coronavirus farm aid: sources. Available	604
online: https://www.reuters.com/article/us-usa-coronavirus-usda-farmaid-idUSKCN21W01K (accessed on 19 May 2021).	605
167. Reuters (2020y). JBS shuts U.S. pork plant, Cargill idles Canadian beef plant on coronavirus. Available online:	606
https://www.reuters.com/article/us-health-coronavirus-usa-meatpacking-idUSKBN2222GN (accessed on 19 May 2021).	607
168. Reuters (2020z). Coronavirus slams West Africa's cashew market, some crops left to rot. Available online:	608
https://www.reuters.com/article/us-health-coronavirus-africa-cashew-idUSKBN23J15R (accessed on 19 May 2021).	609
169. Reuters (2020aa). U.S. food banks run short on staples as hunger soars. Available online: https://www.reuters.com/arti-	610
cle/us-health-coronavirus-foodbanks-insight-idUSKCN2261AY (accessed on 19 May 2021).	611
170. Reuters (2020ab). Britain tells shoppers food is plentiful but supermarkets fret about next week. Available online:	612
https://www.reuters.com/article/us-health-coronavirus-britain-supplies-idUSKBN28W12V (accessed on 19 May 2021).	613
171. Reuters (2020ac). UPDATE 1-Philippines' farm minister fears sugar price spike as coronavirus shuts mills. Available	614
online: https://www.reuters.com/article/health-coronavirus-philippines-sugar-idUSL3N2C32KD (accessed on 19 May	615
2021).	616
172. Reuters (2020ad). World Bank tells G20: Pandemic threatens food security of poor nations. Available online:	617
https://www.reuters.com/article/health-coronavirus-food-worldbank-idINKCN2232LQ (accessed on 19 May 2021).	618
173. Reuters (2020ae). Turkish fruit, vegetable farmers wary as closed borders slice prices. Available online: https://www.reu-	619
ters.com/article/us-health-coronavirus-turkey-crops-idUSKCN21Y258 (accessed on 19 May 2021).	620
174. Reuters (2020af). Fresh produce in Europe set to be more scarce as coronavirus strikes. Available online: https://www.reu-	621
ters.com/article/us-health-coronavirus-produce-insight-idUSKBN21D12V (accessed on 19 May 2021).	622

175. Reuters (2020ag). North America's hog farmers face prolonged pain as pandemic gives way to glut. Available online:	623
https://www.reuters.com/article/us-canada-hogs-idUSKBN25N2IJ (accessed on 19 May 2021).	624
176. Reuters (2020ah). Tesco demands supplier price cuts by July 10 – source. Available online: https://www.reuters.com/arti-	625
<u>cle/uk-tesco-suppliers-idUKKBN24422F</u> (accessed on 19 May 2021).	626
177. Reuters (2020ai). As pandemic empties traditional Uzbek bazaars, supermarkets step in. Available online:	627
https://www.reuters.com/article/us-health-coronavirus-uzbekistan-retail-idUSKBN22C1SU (accessed on 19 May 2021).	628
178. Reuters (2020aj). Onion, garlic get the chop in Indonesia as coronavirus hits imports. Available online: https://www.reu-	629
ters.com/article/health-coronavirus-indonesia-food-idUSL4N2B52MJ (accessed on 19 May 2021).	630
179. Reuters (2020ak). HelloFresh shares at record high on coronavirus sales boost. Available online: https://www.reu-	631
ters.com/article/us-hellofresh-shares-idUSKBN21I0V3 (accessed on 19 May 2021).	632
180. Reuters (2020al). U.S. producer prices accelerate, but overall inflation trending softer. Available online: https://www.reu-	633
ters.com/article/us-usa-economy-inflation-idUSKBN26Z1TR (accessed on 19 May 2021).	634
181. Reuters (2020am). Coronavirus lockdown destroys jobs at record pace in 'paralysed' Spain. Available online:	635
https://www.reuters.com/article/us-spain-economy-unemployment-idUKKBN21K0UH (accessed on 19 May 2021).	636
182. Reuters (2020an). Americans rush to stock up on essentials, retailers scramble to keep up. Available online:	637
https://www.reuters.com/article/us-health-coronavirus-usa-shoppers-idUSKBN21034A (accessed on 19 May 2021).	638
183. Reuters (2021a). USDA devotes \$6 billion to help small, disadvantaged farmers hurt by pandemic. Available online:	639
https://www.reuters.com/article/us-health-coronavirus-usda-aid-idUSKBN2BG2XW (accessed on 17 May 2021).	640
184. Reuters (2021b). Indian rice shipping delays come at bad time as consumer demand spikes. Available online:	641
https://www.reuters.com/article/global-rice-supplies-idUSL4N2JG1EP (accessed on 17 May 2021).	642
185. Reuters (2021c). In a first, most North American robots last year didn't go to automotive plants. Available online:	643
https://www.reuters.com/article/us-usa-economy-robots-idUKKBN29X1DQ (accessed on 19 May 2021).	644
186. Reuters (2021d). U.S. consumer spending decreases further; inflation creeping up. Available online: https://www.reu-	645
ters.com/business/us-consumer-spending-decreases-further-inflation-creeping-up-2021-01-29/ (accessed on 19 May 2021).	646

SC Link	Impact Category	Impacts [References]	Policy Category	Policies [References]
				Absenteeism monitoring [52]
			147 1 6 1	Risk monitoring [91]
			workforce assessment and	Multiple sourcing [86]
			monitoring	Information management and policy coordination with governments [12]
		Absenteeism, unavailability or quarantine, or		Guidelines on seasonal workers and migrant workers [10,101]
		refusal to work in unsafe conditions		Temporary workers [65,104]
Supply side	Uuman	[1,28,29,30, 39,43,48,41,51,26,52,26,8,76,55,79,61,8	Workforce adjustment	Remote workforce training [105]
logistics and	Rosourcos	0,81,82,83,92,84,69,85,86,32,87,89,90,91,77,94,86,6		Optimizing the familial or community workforce [73]
infrastructure	constraints	5,73,96,63,97,98,67,99,100,101,102,40,56,103,104,4		Mechanization and automation of activities with technology [8,38,64,87,107]
management	constraints	6,47,105, 131,132,135,145,149,153,154,158,156,161,		Digital agronomic trainings and information through programs facilitated
and operation		174,175,181,182,185]	Activities automation	collaboratively with government [64]
				Computer-based statistical models [87]
				Greater reliance on automation [103]
				New health and Safety Protocols [61,63,65,69,86,87,90,91,104,107,109]
			Health and Safety Protocols	Use to personal protective equipment and sanitary measures [14,55,87,103,
				107,109]
		Price instability		New practices for food safety and social responsibility [16]
		[4,7,10,11,12,13,14,25, 28,29,31, 32,33,34,39,40,41,4		Guidelines against price gouging [99]
		3,46,49,53,55,58,63,64,65,66,67,69,69,72,	Stability and business continuity	Guarantee the supplementation of food in distribution centers [110]
	SC instability	73,76,77,82,83,87,89,90,93,94,96,97,99,101,102,103	plans	Strong, focused country leadership [75]
Supply-side;		,106,107,109,111,110,112,113,114,115,116,117,118,		Food price monitoring schemes [58,63,69,99]
logistics and		119, 130131,139,152,158,162,163,30,164,168,169,17		Multiple sourcing [86]
infrastructure;		1,173,175,178,180,186]		Government food prices surveillance; price monitor [10,76]

Appendix A – Impacts of pandemics and epidemics and respective mitigation policies on FSCs

management				Direct assistance to farmers [89]
and operation		Government financial assistance for	Crop insurance subsidies [109]	
			companies	Farm support payments [109]
				Food price subsidies [64]
				Information management and policy coordination with governments [12]
			Cooperation and collaboration	Public-private cooperation [43]
				Adjusting marketing strategies [61]
			Stability and business continuity	Strengthening their positions in retail and online markets [90,104]
Supply-side;			plans	Diversification of product lines to fill alternative markets [38]
logistics and		Sales fluctuation [11,55,38,40,43,61,64,65,81,89,97,101,120,122 ,149, 153,168] SC instability	Government financial assistance for companies	Reduction of fees and taxes [49]
management	SC instability		Online in freestwater was	Food delivery app and logistics company [77]
and operation			Online infrastructures	Online platforms [90,104,121]
			Cooperation and collaboration	Link farmers and restaurants directly to food banks [41,104]
			Decision and communication	Information and communication availability [36]
Sumply side		Inventory instability; restocking become harder; storage difficulties [32 , 45, 64,82,100,106 ,131,153,164, SC instability 169,183]	support tools	Direct payments for inventory were made to cattle producers [89]
logistics and			Stability and business continuity plans	Operational and contingency plans [51,129]
infrastructure:				Manage both operation and inventory [82]
management				Synchronize strategic processes [27]
and operation	SC instability			Policy support for infrastructure development such as storage, cold chains
and operation				and pack houses [94]
			Cooperation and collaboration	Private or government institutions need to invest in storage centres [107]
Supply-side;			Stability and business continuity	Limit exposure to the international market and meeting local demand [88]
demand-side;		Order's cancelations and rescheduling	plans	Additional safety stock [150]
logistics and	SC instability	instability [10,13,55,61,64,85,87,89,94,103,		
infrastructure;			Cooperation and collaboration	Collaboration with suppliers and deadline agreement [176]
management			cosperation and condoration	
and operation				

				Local produce/ slow food [83,126]
				Strengthening consumer demand [127]
				Alternative materials and to source local substitutes [87]
			Stability and business continuity	Compensation measures [121]
			plans	Regular training for their employees [65]
		Dre de stien anderen de service (and de stien ante)		Innovative systems (e.g., aquaponics and aeroponics) [53,57]
Supply-side;		Production volume changes (production rate)		Policy coordination with governments and development of new
logistics and		production)	Cooperation and collaboration	manufacturing strategies [12]
infrastructure;	SC instability	[7 10 12 28 30 38 46 49 53 55 57 58 64 65 73 81 83		Supply chain partnership [87]
management and operation		[7,10,12,28,30,38,46,49,53,55,57,58,64,65,73,81,83, 86,95,97,100,103,104,111,116,123,126,127,144,160, 164,183,177,182]	Decision and communication support tools	Integrated governance of food systems at regional Level [83]
				Simplifying the use of financing instruments in rural development programs
			Covernment financial assistance for	[101]
			companies	Support farmers with output-based and credit-based (interest subsidy)
				subsidy [63 ,164,183]
				Flexibility for companies [144]
			Online infrastructures	Accelerating the development of e-commerce [65]
		Fluctuation in supply and demand [5,7,8,14,27,28,29,30,32,36,39,40, 43,46,47,57, 59,61,63,64,65,66,69,72,77,80,81,82, 84,86,87,89,90,93,94,95,96,97,101,104,107,108, 111,113,116,119,123,126,130,140,145,152,156,156, 158,169,170,180]		Synchronize strategic processes [27]
				Strengthening their positions in retail and online markets [90]
				Operational and contingency plans [51,129]
Supply-side:				Mitigation and prevention measures [26,51,108]
demand-side			Stability and business continuity	Declaring minimum support price or fair price [88]
management	SC instability		plans	Promotion of short food chains and local products [65,86,96,101,130]
and operation				More diversified production and supply networks [57]
				Alternative suppliers and distributors [61]
				Inventory policies [87]
				Strategic autonomy [156]
				Information and communication availability [36]

			Decision and communication	Ensure the smooth connection of supply-and-demand information [121]
			support tools	Demand analysis thought analytic and digital techniques [61,107]
			Connection and collaboration	Visibility and coordination among FSCs [86]
			Cooperation and conadoration	Cooperative organizations with local producers [84]
			Online infrastructures	Online ordering and home delivery [86]
Supply-side;				Operational and contingency plans [51,129]
management	SC instability	Iso 421	stability and business continuity	Business continuity plan [91]
and operation		[39,43]	plans	Mitigation and prevention measures [26,51,108]
				First Expired, First Out method [44]
		Reduction of return on investment and		Partial sacrifice of remuneration agreed to by employees and employers [44]
Supply-side;	T 1 1	reduction of the industry's contributions to		Extended office time after the crisis period [44]
management	Financial and	GDP momic Instraints [44]	Economic and reduce costs plans	Minimize trade promotion costs [44]
and operation	constraints			Utilize weekends and holidays as working days [44]
	constraints			Product rotation in trade [44]
				Focus on building online infrastructures [44]
Supply-side;	Financial and	Shortage of working capital and delay in		Minimize trade promotion costs [44]
and operation	constraints	[44]	_	Focus on building online infrastructures [44]
		nancial and		Change of production focus (e.g., toward production of hand sanitizer or
				similar) [55]
Supply-side;			Economic and reduce costs plans	Reshape strategy for business continuity through stress-testing to evaluate
logistics and	Financial and			short-term liquidity and risks [105]
infrastructure;	economic	[14 36 38 39 40 41 45 55 81 82 89		Downsizing their workforce in order to reduce operational costs [55]
management	constraints 94,9	ronstraints 94,97, 149,152,180]		Cost reduction and elimination of non-essential assets [152]
and operation				Remote support centers [45]
				Crisis committee creation [45]
				Economic aid in the form of so-called "relief packages" [97,105]

			Government awereness of economic impacts	Reduction of the turnover tax rate for all small and medium enterprises (SMEs) [49] Effective financing for companies [75,108]
			Decision and communication	Decision Support System (DSS) [62]
			support tools	Smart packing technology [8]
Supply-side;		Transport disruptions; transport restrictions;	Government financial assistance for companies	Government's assistance [10,63,64,97 ,153]
logistics and	Logistics	delays in transportation	Online infractructures	E-commerce between farmers with consumers [69,104,107]
infrastructure;	disruptions	[1,0,10,10,13,13,20,20,29,30,33,30,39,40,41,40,47,40,47 0 52 55 58 61 64 67 68 60 60 76 70 81 85 86 07 00 1	Online Infrastructures	Delivery through communication applications [104,106]
management	uisrupuons	9,52,55,50,01,04,07,06,05,05,77,07,75,01,05,00,77,55,1 00 101 102 103 104 106 111 131 138 146 156 161 1		Policy actions that ensure minimum transportation bottlenecks [99]
and operation		73 174 183]	Transportation and distribution	Opening 'green lane' border crossings to all freight vehicles [101]
		75,174,105]	plans	Alternative transportation modes [86]
				Ships and trains can also be used for fresh foods [86]
				Local suppliers [87,99]
		Restrictions of movement of goods (e.g., border restrictions; export restrictions; import restrictions) Iogistics disruptions 101,103,111,113,114,116,140,132,131,131,137,		Local suppliers and local production [87,137]
				Kep trade open [30,104]
Supply-side;			Transportation and distribution – plans	Salles the products in the domestic market [103]
	Logistics disruptions			Keep major exporters from enacting trade restrictions [98]
infrastructure				More diversified production and supply networks [57,104]
management			Decision support tools	Increasing trade transparency and traceability [88]
and operation				Digital technologies to monitor harvest [172]
°F		141,142,143,145,147,148,152,155,156,	Covernment financial assistance for	Relaxation of laws and impositions [104,141,180]
		164,172,173,174,177,178]	companies	"Green lanes" [174]
			companies	Improve community groups to distribute food [172]
Supply-side:		Food distribution disruption	Decision support tools	Decision Support System (DSS) [3]
logistics and	Logistics disruptions	[1,13,32,39,41,43,49,50,54,55,56,58,	Online infrastructures	E-delivery services; online platforms [8,106]
infrastructure;		61,64,65,69,77,80,81,90,95,97,110,114,	Transportation and distribution	Short-term incentives for distributors [44]
		116,127,128, 145]	plans	Distribution planning and resource allocation [14]

management				Strengthening food storage and inter- and intra-island transport [127]
and operation				Distribution strategy [50]
				Government humanitarian assistance [40,114]
				Reduction of fees and taxes [49,69,103]
		Shutdown of processing plants; closure of	Government financial assistance for	Government purchasing agricultural products from small farmers and shorter
Supply-side;		businesses (momentarily or not)	companies	supply chains [69]
logistics and	Infrastructure	[10,13,14,28,29,30,33,40,41,47,63,68,		Emergency cashflow and liquidity measures [105]
infrastructure;	disruptions	69,80,82,83,89,90,91,95,96,100,101,		Significant funds [40]
management		103,104,105,107,114,116,119,123,127,131132,138,1		Activate the agricultural crisis reserve [101]
and operation		49,151,152,153,156,160,161,166,167,171,173,173,1	Restructuring plans	Divert products to other units [160]
		81]		Market information [94
			Online infrastructures	E-delivery services; online platforms [8,55,65,77,100,104,106,124]
		nfrastructure Constraints on storage space capacity isruptions [14,40,46,61,97,99,126]	Workforce adjustment	Relocating healthy staff from closed plants [126]
			Restructuring plans	Better warehousing infrastructure [99]
				Policy support for infrastructure development [94]
Logistics and				Market support measures [101]
infrastructures	Infractionation			Increasing hours of operation [126]
management	disruptions			Upskilling staff [61]
management	disruptions			Strong focused country leadership [75]
and operation				New operating procedures [50]
			Online infrastructures	Focus on building online infrastructures [44]
			Government financial assistance for	Short term incentives for distributors [44]
			companies	
Supply-side;			Communication technologies	Information and communication availability [36]
logistics and	Relationships	Relationships between stakeholders [62]		Improve communication skills [38]
infrastructure;	between		Conception and collaboration	Sharing of data and information [100,107]
management	stakeholders			Collaboration between stakeholders [107]
and operation				Coordination, cooperation, and support among stakeholders [63]

				Government supported scaling up digital information sharing models [106]
Supply-side; logistics and infrastructure;	Relationships between	ionships een [30, 44,67 ,168] holders	Cooperation and collaboration	Effective partnerships and integration [27,75]
management and operation	stakeholders			Product rotation in trade [44]
Supply-side; logistics and	Difficulty	Difficulty accessing services and materials	Government financial assistance for companies	Government subsidies [100]
infrastructure; management and operation	services and materials		Contingency plans	Assistance manuals and guidelines [64]
	Difficulty accessing	fficulty Input's supply (ex: seeds, fertilizer, and agrochemicals) ressing [8,10,28,33,38,39,40,41,43,48,49, rvices and aterials 55,63,64,65,69,76,77,79,80,82,87, 92,94,97,100,102,107,110,114,120]	Contingency plans	Diversify the supply chain [39,55] Local inputs [100] Establish and operationalize the food bank, community seed bank, and buffer
				stock of fertilizers [63] Increase inventory levels [86]
Supply-side:				Diversify the supply chain [121]
logistics and				Sustaining the critical agricultural inputs [86]
infrastructure;				Anticipated reducing product lines and/or ingredients used [55]
management	materials			Public-private cooperation [43]
and operation			Cooperation and collaboration	Strength public-private-community partnership mechanisms [63]
				Seed and other input supplies should be ensured (government)
			Covernment financial assistance for	[10,48,100,114]
			companies	Provide seeds and fertilizer to farmers in several countries [64]
				Contractual grain production agreements with farmer groups (advanced seed credit to farmers) [64]

				Subsidy programs and relief packages [28,64]
				Government supported programs [64]
				Operational and contingency plans [51,129]
0 1 1		Change in consumer behavior (food		Synchronize strategic processes [27]
Supply-side;		consumption patterns)		Food surveillance measures should include consumer-eating habits [47]
demand-side;	Consumer		Demand analysis plans	Innovative marketing and planning of the distribution [84]
management	benavior	65,66,68,69,76,82,83,84,86,87,89,90,91,94,		Relaxation of competition laws [136]
and operation		95,101,104,105,107,112,116,117,119,124,		Adjust payment methods to consumers' demands [84]
		131,130,177,100]	Information sharing tools	Provide detailed information (e.g., webinars and podcasts) [104]
		Panic buying; panic of food shortages		Efficient information sharing mechanisms [83]
Demand-side; management	Consumer behavior	[7,8,14, 29,30,32,39,41,47,53,56,58,66,68, onsumer havior (9,69,73,77,82,83,86,90,93,94,95,98,99,101,103,104) ,106,109,113,117,121,123,124,125,126,147,30164,1 69,182]	Information sharing tools	Publicly available food supply information [86,96,100]
and operation				Transparent dissemination of information [10,124]
			Social support programs	Measures as social assistance and incentive [67,102]
				Social measures (e.g., donations, increase income) [30,57,72,79,97]
		income changes)		Use of food banks (food at a nominal price) [56,109]
	Income changes) [7,13,39,41,55,56,57,58,63,64,65,68,69, Unemployme 72,73,75,76,77,80,83,86,90,91,92,94,97,99, nt and reduce 101,102,106,108,111,112,114,118,120, income 142,158		Maximize food consumption with minimum wastage [112]	
Demand-side		Linemployme 72 72 75 76 77 80 82 86 90 91 92 94 97 99	Food security plans	A subsidy for selected food products to restrict pricing [86]
		and reduce 101 102 106 108 111 112 114 118 120		Governmental and non-government agencies partnership [86]
		incomo 142 158]		Government supported programs [64,77,94,158]
	liteonie	112,100]	covernment intaricial assistance for	Tax relief according with the gross monthly income [49]
			population	Facilitate access to food and assist the population [112]
		Job losses; employee layoffs; unemployment	Government financial assistance for population	Social insurance (e.g., unemployment benefits) [90]
Demand-side	Unemployme	Inemployme [13,32,38,41,47,49,55,57,58,64,65,69,72,73,76,77,8] t and reduce 5,90,95,96,97,100,108,116,	Social support programs	Food banks and other sources of food assistance [101]
	nt and reduce			Social support programs [57,77,106]
		income 123,127,142,169,181]	Workforce adjustment	Enabling remote working and learning [47,105]